Author’s response to reviews

Title: A case report of Coccidioides Posadasii meningoencephalitis in an immunocompetent host

Authors:

Raynell Lang (raynelllang@gmail.com)
William Stokes (William.Stokes@albertahealthservices.ca)
Jane Lemaire (Jane.Lemaire@albertahealthservices.ca)
Andrew Johnson (Andrew.Johnson@albertahealthservices.ca)
John Conly (John.Conly@cls.ab.ca)

Version: 2 Date: 25 Jul 2019

Author’s response to reviews:

Editor Comments:

1. Please combine your "Discussion" and "Conclusions" sections into one section entitled "Discussion and Conclusions".

Thank you, this has been changed.

2. In the “Ethics approval and consent to participate” section of the Declarations please state “Not Applicable”.

Thank you, I have edited this section.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Thank you.